Author's response to reviews

Title: Avian influenza risk perceptions among Canadian First Nations subsistence hunters: Implications for influenza pandemic plans

Authors:

Nadia A. Charania (ncharani@uwaterloo.ca)
Ian D. Martin (ianmartin@mac.com)
Eric N. Liberda (eric.liberda@ryerson.ca)
Richard Meldrum (meldrum@ryerson.ca)
Leonard J.S. Tsuji (leonard.tsuji@utoronto.ca)

Version: 2 Date: 8 July 2014

Author's response to reviews:

The attached revised manuscript includes line and page numbers as requested.

Thank you.